ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

von Hagen, Jürgen; Dinger, Valeriya

Working Paper Banking sector (under?) development in Central and Eastern Europe

ZEI Working Paper, No. B 06-2005

Provided in Cooperation with: ZEI - Center for European Integration Studies, University of Bonn

Suggested Citation: von Hagen, Jürgen; Dinger, Valeriya (2005) : Banking sector (under?) development in Central and Eastern Europe, ZEI Working Paper, No. B 06-2005, Rheinische Friedrich-Wilhelms-Universität Bonn, Zentrum für Europäische Integrationsforschung (ZEI), Bonn

This Version is available at: https://hdl.handle.net/10419/39448

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Zentrum für Europäische Integrationsforschung Center for European Integration Studies Rheinische Friedrich-Wilhelms-Universität Bonn



Jürgen von Hagen and Valeriya Dinger

Banking Sector (Under?) Development in Central and Eastern Europe Working 0

ISSN 1436-6053



Zentrum für Europäische Integrationsforschung Center for European Integration Studies Rheinische Friedrich-Wilhelms-Universität Bonn

Walter-Flex-Straße 3 D-53113 Bonn Germany Tel.: +49-228-73-9218 Fax: +49-228-73-1809 http://www.zei.de B 06 2005

Banking Sector (Under?)Development in Central and Eastern Europe

Jürgen von Hagen^{*} and Valeriya Dinger^{**}

This Version: February 2005

Abstract:

By introducing a new measure of the banking systems' size, the paper challenges the existing consensus on severe underdevelopment of the CEE banking sectors. We argue that the existing studies on the size of CEE banking systems exaggerate the real degree of underdevelopment because common measures of the size of the banking system produce downward biased results when applied to transition economies. We compare various measures of the size of the CEE banking sectors with those of several "old" European Union (EU) member countries which are used as benchmarks. The comparison indicates that indeed the banking sectors in the CEE countries lag behind the most developed financial systems in the EU, but are very close to the levels in the financially less developed EU countries.

Key words: financial intermediation, transition economies, banking sectors' size

JEL: G21, P34

^{*} Center for European Integration Studies, University of Bonn, Indiana University and CEPR. E-mail address: vonhagen@uni-bonn.de

^{**} Center for European Integration Studies, University of Bonn. E-mail address: valeriya.dinger@uni-bonn.de

1. Introduction

The existence of developed financial institutions is an important prerequisite for an efficient allocation of capital and sustainable economic growth. Therefore, viable and well functioning banking sectors are important for the development path of transition economies where the process of financial intermediation had been profoundly reshaped in accordance with the liberalizing of the business environment. The literature on banking in transition economies argues that banking sectors in Central and Eastern European countries are much smaller than those in mature market economies (Bonin and Wachtel, 2002; Anderson and Kegels, 1998), and that this puts a limit on the growth potential of these economies.

In this paper we present an analysis of the size of the banking sectors in ten Central and Eastern European (CEE) countries¹. We focus especially on the sectors' ability to provide efficient financial intermediation between savers and investors in the economy. By introducing a new measure of the size of banking systems, we challenge the existing consensus on the severe underdevelopment of the CEE banking sectors. We argue that common measures of the size of the banking system produce downward biased results when applied to countries with a low stock of financial wealth in general and to transition economies in particular. Existing studies on the size of CEE banking systems, therefore, exaggerate the real degree of underdevelopment. We compare various measures of the size of the CEE banking sectors with those of several "old" European Union (EU) member² countries which are used as benchmarks. The comparison indicates that indeed the banking sectors in the CEE countries lag behind the most developed EU countries.

Next, we explore various reasons for the comparatively low levels of financial intermediation as measured by traditional methods used in the literature (Bonin and Wachtel, 2002; Anderson and Kegels, 1998). We concentrate in detail on three such arguments. The first points to the low initial levels of bank activity in CEE countries as a cause of the low current

¹ Bulgaria, the Czech Republic, Estonia, Hungary, Latvia, Lithuania, Poland, Romania, Slovakia and Slovenia

² later on only referred to as EU countries

levels of the banks' financial intermediation. The second maintains that as capital markets develop they at least partially substitute for the necessity of financial intermediation by banks. Finally, the third argument stresses the low involvement of individual CEE banks in financial intermediation as a reason for the lower financial intermediation performance of the banking sectors.

We focus this analysis on the developments in ten CEE countries: Bulgaria, the Czech Republic, Estonia, Hungary, Latvia, Lithuania, Poland, Romania, Slovakia and Slovenia during the time period from 1993 to 2001.

Regarding the size of the banking industries, we relate the level of financial intermediation to proxies for the available financial wealth in the sample countries, rather than to GDP as the literature commonly does. We do this to account for the fact that transition economies have lower stock of private financial wealth than mature market economies³. With low wealth levels the size of the banking industry (financial intermediation in general) is constrained by the low stock of intermediable resources, unless capital inflows compensate fully for the small volume of domestic financial wealth. The literature on the integration of capital markets (Bayoumi, 1990; French and Poterba, 1991; Baxter and Crucini, 1993)⁴ suggests that this is typically not the case.

Due to the low stock of financial wealth, banking sectors appear small relative to GDP even though their size relative to financial wealth may be comparable with the size of banking sectors in mature market economies. Indeed, the results of our analysis point to the fact that GDP-based measures of the size of CEE banking sectors underestimate the ability of the banking sector to intermediate financial wealth. When the financial intermediation ability of the banking sectors is measured relative to financial wealth, the gap between the CEE countries and the EU countries is smaller than indicated by GDP-based measures.

³ The fact that the stock of financial wealth in CEE transition economies is low has been pointed out by the World Bank and IMF (see Koch-Weser, 1996). Unfortunately there exist, to our knowledge, no systematic studies on the stock of financial wealth in CEE.

⁴ This literature concentrates on analyzing the 'home-bias puzzle' that was initially pointed out by Feldstein and Horioka (1980)

Nevertheless, the gap still exists and the banks' financial intermediation in all CEE sample countries is underdeveloped relative to EU countries with developed financial systems.

This paper also explores alternative arguments, presented in the literature, explaining this low level of financial intermediation provided by the banks.

The first of these arguments is based on the fact that the initial level of bank intermediation at the onset of the transition was extremely low (see Anderson and Kegels, 1998). The underdevelopment of bank intermediation in planned economies is explained by the fact that the intermediation function was generally performed by the governments. We test this argument by constructing two measures of marginal intermediation that illustrate the changing importance of bank intermediation. The first measure compares the change of bank deposits to gross savings in the economy; the second ties the change of domestic credit to the sum of investments and government deficit. These two measures allow us to illustrate the degree of bank penetration into the financial intermediation chain. Generally, the results of our analysis indicate an increasing importance of both deposit accumulation and loan supply by banks in the early transition period. Later in the transition process a trend of increasing importance of bank intermediation is only observed with regard to deposit gathering. Deposit accumulation by CEE banks is therefore approaching the levels prevailing in the EU. With regard to loan supply, the trend of increasing bank importance slowed during the late 1990s. Therefore, our analysis only partly supports the argument that the low starting levels of bank intermediation can explain the low level of bank intermediation still observed in CEE countries.

Secondly, we focus on the argument that the development of capital markets has evoked changes in the financial intermediation function of banks. This view is thoroughly discussed in Danthine et. al. (1999). Danthine et. al. (1999) argue that this could either be because the capital markets substitute the banks in undertaking the main share of financial intermediation between savers and investors in the economy, or because the banks' financial intermediation has changed in a way that the lending activity is moving towards securitization of loans and therefore is reflected off-balance sheet. As suggested by Fender and von Hagen (1998),

securitization of loans does change the behavior of banks and may modify monetary policy transmission. However, since the process of securitization does not eliminate a bank's role in monitoring borrowers, the banks continue to play an essential role in the intermediation chain, even though it is no longer reflected in the existence of loans on the banks' books. Nevertheless, these explanations are based on the assumption of sufficient capital market development. We present a closer look at the development of capital markets in CEE in absolute measures and relative to banking sector development in order to see whether these are the valid explanations for the low levels of bank intermediation in CEE countries. The results show that CEE capital markets are severely underdeveloped, even relative to banking sectors. Thus, the argument that the insufficient levels of classical financial intermediation by CEE banks are caused by the advanced development of capital markets lacks empirical support.

Finally, we explore the argument that the level of bank intermediation in CEE countries is low because the financial intermediation productivity of banks is low as measured by aggregate level variables, such as loans to total banking industry assets and deposits to total banking industry assets (Miller and Petranov, 2002). To test this argument, we analyze micro level balance sheet data from a large sample of CEE banks. The use of micro level data is an important innovation in our research⁵. It allows us to detect heterogeneity among the banks in individual banking systems and indicates that the low aggregate levels of financial intermediation productivity in some CEE countries reflect the specialization in deposit gathering by some banks and in loan supply by others.

Throughout this analysis we use the three EU countries Germany, the Netherlands and Greece as benchmarks. We prefer a comparison with these benchmark countries over average variables for the whole sample of EU members as the differences of the structure of the financial systems across the EU countries are significant⁶ and distort the overall comparison.

⁵ To our knowledge the existing literature of financial intermediation activities of banks in transition economies only focuses on aggregated data

⁶ For detailed discussions on the differences across European banking systems see Allen and Gale (1999), Demirguc-Kunt and Levine (2001), Danthine, et al (1999)

Choosing Germany, the Netherlands and Greece as benchmark countries, we have representatives of three different types of financial structures coexisting in the Union⁷. First, Germany being the largest banking system of the EU is a textbook example of highly developed bank-based financial system (similar to Italy, Austria, Belgium, etc.). A comparison of the CEE banking sectors with that of Germany will give us an idea of the extent to which the financial intermediation activities of the CEE banks are approaching those of the banking institutions in developed bank-based systems. The Netherlands (similar to UK, Denmark and Sweden) typify a developed market-based financial system⁸. We can expect that Dutch banks have a different intensity of involvement in classical financial intermediation relative to bank-based systems since in countries with market-based financial systems the function of channeling funds to the real economy is shared between the banking institutions and the equity and debt markets. The inclusion of the Netherlands as a benchmark case controls for the possibility that some of the CEE countries might have directed the transformation of their financial systems towards a more market-based system. Finally Greece (a country where both the banking system and the capital markets⁹ are considered underdeveloped) serves as an example of a relatively underdeveloped bank-based financial system.

The analysis is based on country level data provided by the International Financial Statistics (IFS) issues of the International Monetary Fund, the European Bank for Reconstruction and Development's Transition Reports, Thomson Financial's Datastream, International Statistics issues of the Bank for International Settlements and on bank level data provided by BankScope^{10,11}.

⁷ Demirguc-Kunt and Levine (2001)

⁸ Demirguc-Kunt and Levine (2001) define a financial system as market-based or bank-based depending on an index consisting of the following ratios: market capitalization versus bank assets, total value traded on the security market versus bank credit and security market turnover versus overhead costs of the banking sector.

⁹ Demirguc-Kunt and Levine (2001)

¹⁰ BankScope is a database provided by Bureau van Dijk and FITCH.

¹¹ Table A.1 in Appendix A presents the data sources for each variable.

The rest of the paper is organized as follows; Section 2 presents alternative measures of the size of the banking sectors and compares these measures across the CEE and the EU benchmark countries. Section 3 focuses on the changes in the importance of bank intermediation. Section 4 analyses the development of capital markets in CEE countries relative to banking systems. Section 5 presents the results of the micro-level analysis of bank financial intermediation productivity, and Section 6 gives the conclusion.

2. Are banking sectors in CEE countries smaller than those in the EU?

This section concentrates on different size measures of CEE banking sectors and their ability to provide financial intermediation. First, following the literature¹² we measure the size of the banking industry as the aggregate volume of bank assets in the country relative to GDP. Furthermore, following Demirgüc-Kunt and Levine (2001), we include two additional measures of banking sector performance with regard to financial intermediation. One is the ratio of deposits to GDP, which measures the deposit-gathering function of banks. The other is the ratio of domestic bank credit to GDP, which measures the loan supply function of the banking sectors.

	1993	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	150	133	93	158	43	33	33	34	39
Czech Republic	95	99	103	100	109	103	103	99	96
Estonia	27	31	31	36	55	50	56	61	67
Hungary	67	62	58	57	58	58	59	60	61
Latvia	68	44	31	36	47	42	47	58	66
Lithuania	24	27	22	20	22	24	26	28	29
Poland	47	44	42	44	45	48	50	49	52
Romania	40	31	35	39	26	29	27	25	25
Slovakia	76	65	65	78	97	90	79	86	90
Slovenia	54	59	62	64	67	69	70	71	84
Germany	153	154	160	170	180	191	218	224	226
Greece	92	89	98	99	101	102	112	116	126
Netherlands	170	162	171	181	199	206	240	246	259

Table 1: Aggregate bank assets to GDP, in %

Source: Own calculations based on IFS data

As presented in Table 1 banking sectors in CEE countries are significantly smaller with respect to the ratio of aggregate volume of bank assets to GDP than those of Germany and the Netherlands. Only in the Czech Republic is the size of the banking sector, as measured by this

¹² See Demirgüc-Kunt and Levine (2001) and Bonin (2001)

proxy, close to the lowest ranked EU country (Greece). The rest of the sample transition countries' banking sectors are much smaller than the one of Greece. Furthermore, this measure indicates a trend of increasing the size of the banking sectors only in Estonia and Slovenia. On the contrary, in Bulgaria a drastic decline in the ratio of bank assets to GDP is observed. In the rest of the CEE countries the levels of the variables in 2001 are very similar to the ones in 1993.

	1993	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	68	71	58	64	24	20	21	22	27
Czech Republic	62	62	66	63	63	59	57	64	66
Estonia	19	18	17	20	25	23	27	31	35
Hungary	44	42	40	41	40	39	39	39	40
Latvia	21	24	15	14	17	17	17	20	23
Lithuania	16	18	15	11	12	13	15	17	21
Poland	29	29	28	29	31	34	37	37	40
Romania	18	17	20	23	21	22	21	20	20
Slovakia	56	55	56	58	56	54	56	59	60
Slovenia	28	31	34	37	40	42	43	45	52
Germany	61	60	60	64	64	66	99	98	101
Greece	52	56	57	59	59	55	58	57	98
Netherlands	77	73	75	76	77	86	97	100	104

Table 2: Aggregate deposits to GDP, in %

Source: Own calculations based on IFS data

	1993	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	127	104	71	112	23	18	18	17	20
Czech Republic	73	76	76	72	72	64	60	54	49
Estonia	13	13	14	22	32	33	35	39	43
Hungary	97	93	82	72	66	63	53	55	50
Latvia	18	23	15	13	15	18	20	25	31
Lithuania	16	19	15	12	13	14	17	16	16
Poland	41	37	32	33	34	35	38	34	36
Romania	21	18	24	29	19	22	18	14	12
Slovakia	72	57	48	56	66	62	59	58	62
Slovenia	34	32	37	36	36	40	43	45	48
Germany	99	102	103	109	113	119	121	125	126
Greece	110	101	96	91	88	85	94	101	100
Netherlands	106	107	112	119	127	131	147	153	154

Table 3: Aggregate domestic bank credit to GDP, in %

Source: Own calculations based on IFS data

With regard to the variables measuring the financial intermediation performance of banks relative to GDP, we observe both lower deposit accumulation and loan supply of CEE banks. As shown in Table 2, concerning the deposit accumulation function, the Czech Republic and Slovakia have the most developed banking sectors among the accession countries. But even

there, banks accumulate significantly less deposits relative to GDP, than the banks in the EU benchmark countries.

Table 3 illustrates that the loan supply function of CEE banks is especially underdeveloped. Whereas in Germany and the Netherlands the value of domestic bank credit significantly exceeds the level of GDP, in the Czech Republic, Hungary, Slovakia and Slovenia domestic bank credit only equals about half of GDP. In the rest of the sample countries domestic bank credit represents less than a third of GDP. Furthermore, only in Estonia and Slovenia are the values of domestic bank credit to GDP ratio increasing.

In view of this, we confirm the notion that in CEE countries banking sectors are smaller and provide less financial intermediation relative to GDP than in mature market economies. However, the question arises, whether financial intermediation capacity should be measured in relation to GDP. Financial intermediation is the sum of activities serving the intermediation between savers and investors. It requires the existence of financial resources (wealth) to be intermediated. However this stock of financial wealth is, for historical reasons, lower in transition economies (see Koch-Weser, 1996) than in mature market economies. Thus, the size of the banking industry and its financial intermediation capacity measured in relation to the stock of financial wealth might represent a better measure for the development of the banking sector.

Unfortunately, there exist to our knowledge, no systematic data on the stock of financial wealth in the CEE countries. To close this gap we introduce a measure of the stock of financial assets in the economy as a proxy for financial wealth and then relate the size of the banking sectors to this new financial wealth proxy. To create such a measure we add together broad money (equal to the sum of currency outside the banks and the demand, time, savings and foreign currency deposits of residents), money market instruments, and bonds issued by banks and monetary authorities. In this measure we do not include equity shares in the stock of financial assets, because data is not available for all countries and years. Non-bank corporate debt securities are also not included in this and the following proxy for financial

wealth because, as we show later¹³, government securities held by banks, represent in CEE countries the main share of domestic securities¹⁴.

Table 4 presents the size of the banking sectors relative to this new financial wealth proxy. Moreover, for those years and countries where data on stock market capitalization is available we add the stock market capitalization to the sum of broad money, money market instruments and bonds¹⁵. Table 5 presents the size of the banking sectors relative to this broader measure.

As illustrated in Table 4 banking sectors of the CEE accession countries are generally smaller than the banking sectors in the EU benchmark countries, but the gap between these two groups of countries is much smaller than indicated by the traditional measure. So, for example, Latvia and Slovenia according to this new measure have banking sectors of a similar size to that of Greece. Furthermore, the banking sectors of Bulgaria and Romania, which according to the bank assets to GDP measure were classified as severely underdeveloped, have according to this measure a size comparable to the one of the Czech or the Hungarian banking sectors.

	1993	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	194	170	142	223	133	115	108	96	94
Czech Republic	111	110	106	109	110	107	102	99	105
Estonia	69	83	90	102	127	117	113	120	118
Hungary	107	107	104	103	105	107	110	109	107
Latvia	175	104	107	124	142	139	158	155	162
Lithuania	64	79	78	87	89	89	91	91	92
Poland	115	112	111	110	106	104	102	104	101
Romania	173	140	135	137	106	117	108	106	108
Slovakia	122	106	102	114	148	144	123	127	131
Slovenia	171	169	162	158	151	146	145	143	146
Germany	189	187	194	199	210	215	224	227	228
Greece	105	101	117	123	130	139	133	143	153
Netherlands	139	137	141	146	160	163	171	168	170

Table 4: Aggregate bank assets to a proxy of the stock of aggregate financial assets, in %

Source: Own calculations based on IFS data

¹³ See Section 4

¹⁴ Central banks' data indicate that in the Czech Republic in the period 1995-2001 banks held between 60 and 70% of the volume of domestic securities. In Hungary the share of domestic securities held by banks is about 55% and in Poland around 80%.

¹⁵ The assumption, based on the Feldstein-Horioka home bias puzzle, that equity traded in domestic markets is mostly held by domestic agents is made here. See Folkerts-Landau et. al. (1997) for data on the low volumes of international portfolio investments in CEE.

As shown in Table 5 if banking system assets are related to a broader proxy of domestic wealth including equity shares, then again in most CEE countries the size of the banking systems are very close to that of Greece. Furthermore, Latvian and Slovakian banking sectors have by this measure a size similar to the Dutch banking sector¹⁶. These results indicate that the ratio of bank assets to GDP presents a downward biased measure for the size of the banking industry in the accession countries. The magnitude of the bias is especially large in the poorer CEE economies, where financial wealth is small relative to officially measured economic activity.

Table 5: Aggregate bank assets to a proxy of the stock of aggregate financial assets including stock market capitalization, in %

	1993	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	na	na	141	222	137	91	90	84	82
Czech Republic	na	95	81	81	86	89	83	80	87
Estonia	na	na	na	na	81	96	65	71	71
Hungary	103	100	94	84	64	69	66	74	77
Latvia	na	na	107	113	120	115	131	127	132
Lithuania	na	77	71	58	52	65	66	62	62
Poland	105	103	101	95	87	81	72	74	73
Romania	na	140	133	136	98	103	96	91	96
Slovakia	na	95	92	98	130	134	116	120	123
Slovenia	na	152	154	145	125	116	116	96	105
Germany	168	165	153	138	134	132	129	136	144
Greece	93	88	98	97	92	88	85	88	85
Netherlands	94	89	88	82	79	76	94	101	125

na – data not available

Source: Own calculations based on IFS, Datastream and Transition Report

In sum, banking sectors in CEE countries are generally smaller than those of the developed financial system in the EU countries. However, the gap is not as severe as argued in studies based on the traditional approach of measuring the size of the banking system and the level of bank intermediation in relation to GDP. When we relate the sum of aggregate banking assets to a proxy for the stock of financial assets, the banking sectors in all CEE countries, but Latvia, are indicated to be only slightly smaller than the one of Greece. The Latvian banking sector is even larger than the Greek one as measured in this relation. These results indicate that the low level of bank intermediation in CEE countries is to a large extent caused by the low level of accumulated financial wealth.

¹⁶ The measure presented in Table 5 indicates a much smaller size of the Dutch banking sector than that presented in Table 4, which is due to the market-orientation of the Dutch financial system. In the environment of highly developed capital markets banks intermediate a smaller proportion of financial wealth.

3. Is the role of bank intermediation in the CEE countries increasing?

In this section we focus on the question of whether the role that banks play in the process of channeling funds from savers to investors is gaining in importance. For this purpose we introduce two variables that we call 'measures of marginal financial intermediation'. The first variable equals the ratio of the change in deposits between time period t and t-1 to the gross savings in period t. This ratio measures the change in the relative importance of bank deposits in the stock of financial wealth. However, comparability of these ratios across countries is conditional on the differences among the money multipliers and we should bear in mind these differences when analyzing the values of the above ratio. The second measure equals the ratio of the change of domestic bank credit between time period t and t-1 to the sum of investment and government deficit¹⁷ in period t. This measure is an illustration of the share of investments in the economy that has been financed by bank credit.

Table 6: Marginal financial intermediation: Δ aggregate deposits with banks/gross savings, in %

	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	156	132	54	16	6	17	27	49
Czech Republic	32	43	16	16	6	0	38	30
Estonia	22	22	40	48	7	27	33	35
Hungary	33	38	41	41	27	23	25	29
Latvia	34	-38	13	32	9	6	26	21
Lithuania	41	19	-4	21	13	14	25	26
Poland	52	35	36	30	34	23	21	25
Romania	52	43	46	54	43	38	29	35
Slovakia	31	34	26	13	8	19	27	20
Slovenia	39	34	30	32	28	21	30	51
Germany	6	12	21	8	20	15	7	24
Greece	69	47	43	34	5	39	16	19
Netherlands	1	21	20	23	27	32	38	37

Source: Own calculations based on IFS data

As shown in Table 6, the change of bank deposits represents in CEE countries on average a higher share of gross saving than in the EU benchmark countries. The fact that CEE countries have lower money multipliers than the EU benchmarks¹⁸ reinforces the implication that CEE banks are accumulating a higher share of savings than banks in the EU. Especially in 1994-1996 CEE banks accumulated a very high share of savings. Later on the pace slowed down

¹⁷ Government deficit is included in the denominator since the data for the nominator does not allow us to distinguish between credit to the private sector and credit to the government.

¹⁸ See Central banks' Annual Reports

but the values of this ratio remain higher in the accession than in the "old" member countries. This result indicates that the deposit gathering performance of CEE banking sectors is catching up to the EU levels.

With regard to the loan supply function, at least in the mid 1990s the change in bank loans represented a very large share of investment in all CEE countries except Slovakia (see Table 7). Later on, the importance of bank credit decreased in most of the countries to levels below the ones of the EU benchmarks. In times when substantial amounts of bad loans were written-off the bank books, we even observe negative change in the volume of domestic credit outstanding (i.e. Slovakia in 1994, Latvia in 1995). Only in four of the sample CEE countries (Bulgaria, Estonia, Latvia and Slovenia) the share of bank credit to investments is comparable to any of the EU benchmarks. Therefore, the data present no evidence that the level of bank loan supply in the CEE countries is catching up to the EU levels.

Table 7: Marginal financial intermediation: ∆ domestic bank credit/(investment +government deficit), in %

	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	58	61	45	16	31	42	73	61
Czech Republic	46	35	17	16	-6	-8	-9	-1
Estonia	11	19	43	47	16	12	32	30
Hungary	52	37	21	30	23	-11	33	6
Latvia	64	-34	3	25	16	8	27	28
Lithuania	31	7	1	14	8	9	-2	6
Poland	42	26	34	27	23	23	7	18
Romania	43	45	49	25	55	19	10	13
Slovakia	-15	-3	37	44	5	2	14	23
Slovenia	29	46	19	17	32	27	29	30
Germany	30	21	31	29	41	39	35	33
Greece	19	37	21	27	19	66	63	54
Netherlands	31	44	52	72	70	74	84	45

Source: Own calculations based on IFS data

To summarize, at least in the early transition period, CEE banking sectors were characterized by high levels of marginal intermediation. Whereas this process is still valid with regard to the deposit gathering activities of banks, the later years are marked by no catching up concerning the loan supply by CEE banks. Thus it is not likely that the level of bank loan supply in the CEE countries will reach the level of the EU benchmarks in the future.

4. Are capital markets compensating for the low level of financial intermediation provided by banks?

In this section we explore the argument that the low levels of financial intermediation provided by the CEE banks are a result of the development of capital markets. So, for example, Danthine, et. al. (1999) argue that banks may provide lower levels of classical financial intermediation (as measured by deposit and loan volumes) because capital market development serves at least partially as a substitute. Furthermore, they point out the fact that, even if financial intermediation is mainly performed by banks, classical measures like loans to GDP may fail to reflect this since they do not account for off-balance sheet activities of banks. In this section we present an overview of capital market developments in CEE countries and analyze their relevance for the low levels of loan supply.

]	1993	1994	1995	1996	1997	1998	1999	2000	2001
Czech Republic	all issuers	11	17	23	22	26	36	47	45	43
	government	11	14	18	15	17	29	38	34	34
Hungary	all issuers	23	28	23	31	27	32	32	32	38
	government	23	27	23	30	25	30	30	30	35
Poland	all issuers	22	20	20	16	15	18	16	19	24
	government	21	20	20	16	15	18	16	19	24
Germany	all issuers	69	85	85	75	74	96	86	92	81
	government	22	27	26	24	24	31	29	32	32
Greece	all issuers	66	74	74	81	78	82	71	77	78
	government	62	70	71	80	77	82	71	77	78
Netherlands	all issuers	69	78	79	78	81	88	87	94	94
	government	45	48	50	49	48	51	46	44	41

Table 8: Domestic debt securities to GDF, in 7	Table 8:	Domestic	debt	securities	to	GDP,	in	%
--	----------	----------	------	------------	----	------	----	---

Source: Bank of International Settlement

Following Demirgüc-Kunt and Levine (2001) we use the ratio of stock market capitalization to GDP as a measure for capital market development. This measure ignores both government and corporate debt markets, but is still frequently used in the literature on transition economies (Anderson and Kegels, 1998; Hencsey and Hultgren, 2001) as a proxy for capital market development. Reasons why this measure is still used in the transition literature include the following. First, systematic data about debt markets that would allow a cross-country comparison is unavailable¹⁹. Second, as shown by Pissarides (2001) and Hencsey and Hultgren (2001) corporate debt markets are negligibly small in CEE countries. Hencsey and

¹⁹ The Bank for International Settlement provides debt market data for only three of the CEE countries (the Czech Republic, Hungary and Poland)

Hultgren (2001) mention that among the CEE countries debt (both government and corporate) markets are most developed in the Czech Republic, Hungary and Poland.

However as we show in Table 8 even in these countries the size of non-government security markets are negligible as compared to those in Germany and the Netherlands. Furthermore, most of the government debt securities are held by banks (Bonin, 1998)²⁰ and thus developed government securities markets cannot account for transfer of intermediation from banks to markets.

	1993	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	na	na	1	0	0	8	6	5	6
Czech Republic	na	14	30	31	27	20	23	23	19
Estonia	na	na	na	na	25	9	37	35	38
Hungary	2	4	6	12	35	30	36	26	22
Latvia	na	na	0	3	6	6	6	8	9
Lithuania	na	1	3	11	18	10	11	14	15
Poland	4	4	4	7	10	13	20	19	20
Romania	na	0	0	0	2	3	3	4	3
Slovakia	na	7	7	12	9	5	4	4	5
Slovenia	na	4	2	4	9	12	12	24	22
Germany	24	24	24	29	40	48	72	67	58
Greece	12	14	16	21	32	43	48	51	53
Netherlands	59	64	74	96	128	145	185	170	132

Table 9: Stock market capitalization to GDP, in %

Source: Datastream and Transition Report

Table 9 presents the values of the ratio of stock market capitalization to GDP in the accession and the benchmark countries. The data illustrate the gap between the size of stock markets relative to GDP in the accession countries and the EU benchmarks. Even in Greece, the country that represents an underdeveloped bank-based system, the stock market has a much larger size relative to GDP than the highest developed stock markets in the accession countries, those in Estonia and the Czech Republic.

Furthermore, in the CEE countries we do not observe a general trend of approaching the higher EU levels. In the early transition period the volume of market capitalization increased in all countries due to government efforts to enhance capital market participation through privatization schemes. Later on, in most of the countries the process of capital market

²⁰ See also the discussion in Section 2

development slowed down significantly. Only in Estonia and Poland is a trend of further development observed.

	1993	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	na	na	18543	78970	21736	426	549	676	621
Czech Republic	na	696	343	321	408	512	448	426	500
Estonia	na	na	na	na	224	536	152	173	175
Hungary	2931	1480	995	456	165	194	164	230	273
Latvia	na	na	15616	1206	766	687	789	698	730
Lithuania	na	2655	849	176	125	238	245	200	190
Poland	1259	1246	1086	671	472	363	248	259	261
Romania	na	na	8648	19508	1302	878	860	648	813
Slovakia	na	892	969	678	1044	1913	2076	2204	2008
Slovenia	na	1441	3464	1782	717	563	586	294	376
Germany	649	652	657	588	450	396	301	336	391
Greece	768	647	609	459	314	256	233	227	237
Netherlands	288	254	231	188	156	142	112	122	159

Table 10: Aggregate bank assets to stock market capitalization, , in %

na – data not available

Source: Own calculations based on IFS, Datastream and Transition Report

To judge the relative importance of capital markets in CEE financial systems and to provide evidence on the bank-based orientation of the financial systems, we compute (following the approach of Demirgüc-Kunt and Levine, 2001) the ratio between aggregate bank assets and stock market capitalization for each of the observed countries. Demirgüc-Kunt and Levine (2001) compute this ratio for a sample of 95 countries around the world and classify those countries with values above the median as bank-based and those with values below the median as market-based. The median ratio value in their sample is close to 200%. As shown in Table 10 among the CEE countries only Estonia has developed a market-based financial system according to this classification. In two other countries, Hungary and Lithuania, stock markets do play a significant role at least in the period after 1996. In contrast, in Bulgaria, Latvia and Slovakia, stock markets play only a marginal role compared to the banking sectors. In the remaining sample countries the values of bank assets to stock market capitalization are well below Germany's.

Altogether, the data point to a severe underdevelopment of capital markets in most CEE countries, which implies that banks are still the main providers of financial intermediation. Furthermore, it is unlikely that in financial systems with such small security markets banks could shift lending activity off balance sheet through securitization. These results indicate that

the low level of financial intermediation provided by banks cannot be explained by highly developed capital markets.

5. Is the financial intermediation productivity of CEE banks low?

In this section we extend our analysis with a presentation of micro level data from bank balance sheets. We try to answer the question of whether the low levels of deposit accumulation and loan supply on the aggregate level are results of low involvement of individual banks in classical banking functions namely, deposit gathering and loan supply. To explore this question we concentrate on how banks finance their operations and on the main positions in their investment portfolio. We regard the degree to which a bank is involved in deposit gathering and in loan supply as a bank's financial intermediation productivity. The fact that we concentrate on classical loan and deposit functions when estimating the financial intermediation productivity of banks is motivated by the discussed unavailability of more advanced intermediation channels in CEE (see Section 4).

We use two variables from bank balance sheet as proxies for bank financial intermediation productivity. On the asset side a higher involvement in classical credit activities implies that loans represent a high proportion of assets²¹. Therefore, we choose the ratio of *loans to total assets* as a measure of credit activity. Unfortunately, BankScope does not provide separate entries on bank loans to private and public institutions. We use the entry of "loans", which includes both lending to private and public sector, keeping in mind that this measure overestimates the de facto supply of loans on commercial terms²². On the liability side, deposit gathering activity is defined in the literature as "receiving deposits from the public" or "taking deposits from individuals"²³. When analyzing the deposit accumulating activities we

²¹ The structure of the balance sheet used by BankScope is presented on Figure A.1 in Appendix A

²² The importance of this ratio can be a doubtful indicator of banks' involvement in channeling funds to the real sector if most of the credits are granted to government and municipalities. Unfortunately such information is not provided by BankScope for the Central and Eastern European banks. Nevertheless, we argue that the ratio presents relatively unbiased idea of the banks' involvement in credit supply, since government debt, especially in the years of advanced transition, is securitized (see Miller and Petranov, 2002 and Anderson and Kegels, 1998). The large volumes of government bonds hold by the banks present further evidence for the securitization of government debt.

²³ Freixas and Rochet (1997)

therefore concentrate on customer deposits and disregard inter-bank deposits (see Figure A.1 in the Appendix). Therefore, the ratio of *customer deposits to total deposits*²⁴ can be used as a measure for the involvement of banks in deposit gathering activities. This ratio indicates to what extent a specific banking institution is engaged in gathering deposits from the public.

Table 11 and Table 12 present the average values²⁵ of these ratios for each of the CEE sample countries and the EU benchmarks.

	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	65	38	32	25	25	28	29	31
Czech Republic	52	48	50	49	47	39	33	33
Estonia	40	44	51	51	57	56	57	55
Hungary	36	35	35	37	37	41	49	50
Latvia	31	24	23	26	45	43	37	46
Lithuania	55	56	45	39	41	45	41	41
Poland	29	33	38	41	45	49	47	46
Romania	40	49	52	41	32	28	31	34
Slovakia	49	45	47	45	44	47	43	31
Slovenia	49	43	42	43	48	52	52	48
Germany	60	59	58	57	54	51	49	48
Greece	35	36	38	36	37	38	42	46
Netherlands	67	61	56	55	53	56	54	54

Table 11: Loans to total assets, weighted average for the banking system, in %

Source: Own calculations based on BankScope data

Table 12: (Customer deposits	to total deposits,	weighted averag	e for the banking	g system, in %
-------------	--------------------------	--------------------	-----------------	-------------------	----------------

	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	34	74	66	86	89	93	93	93
Czech Republic	71	70	70	70	73	73	76	78
Estonia	94	82	81	74	78	80	87	92
Hungary	74	78	78	74	75	79	78	80
Latvia	88	92	86	86	77	79	85	83
Lithuania	83	90	85	87	82	84	85	85
Poland	81	81	77	79	79	80	80	83
Romania	62	58	58	71	86	85	88	90
Slovakia	67	74	74	74	73	76	80	89
Slovenia	85	87	88	90	88	86	85	83
Germany	63	61	60	57	58	55	57	59
Greece	88	88	87	85	93	92	91	92
Netherlands	73	72	70	69	67	70	68	68

Source: Own calculations based on BankScope data

As shown in Table 11 in most CEE countries loans represent a smaller share of bank assets than in the developed EU financial systems, Germany and the Netherlands. Exceptions are Estonia and Slovenia (for the whole period), Bulgaria in 1994, the Czech Republic in 1994 to

²⁴ The variables "*loans to total assets*" and "*customer deposits to total deposits*" are also called in the literature indicators for banks' business mix (De Bandt and Davis, 1999) or indicators for banks' output mix (Kwan, 2002).

²⁵ Weighted by the volume of a bank's total assets

1998 and Hungary in 2000 to 2001. These results indicate that the low supply of loans on the aggregate level is a reflection of the low average involvement of the individual banks in lending activities.

The data presented in Table 12 illustrate that CEE banks' performance with regard to deposit gathering is on average comparable to that of the EU benchmarks. Nevertheless, differences exist: the average ratios for Estonia, Latvia, Lithuania and Slovenia are generally higher than the those in the rest of the CEE countries and Germany and the Netherlands.

Various articles on the development of the banking sectors in transition economies have pointed out that the population of banks in many CEE countries is characterized by a high level of heterogeneity. So, for example, incumbent saving banks are said to have a different involvement in deposit gathering and loan supply activities than newly entering private banks (Bonin, 1998). Furthermore, foreign banks take on different banking activities than domestic banks (Clarke, et. al, 2003). In the presence of heterogeneity the degree of financial intermediation provided by CEE banks is better described by the distribution of the variables measuring financial intermediation productivity rather than by aggregating these variables.

Figures A.2 to A.14 in the Appendix illustrate the distribution of the variable *loans to total assets* for the CEE countries and the EU benchmarks in the period 1993-2001. The figures show significant differences among the EU benchmarks. This confirms our initial prior that working with benchmark cases is superior to using an EU-wide average. Generally German banks show the highest involvement in lending; almost 75% of German banks invest more than 60% of their assets in loans. Most Greek banks, on the contrary, concentrate only a small share (below 40%) of their assets in loans. The distribution of Dutch banks indicates significant differences across banks: about 20% of banks of banks invest almost exclusively in loans, whereas another 20% hold only a minor share of loans in their assets.

The comparison between the CEE countries and the EU benchmark countries shows that in all CEE countries the share of banks that are heavily involved in lending is smaller than in the developed EU financial systems, Germany and the Netherlands. Only a minor proportion of

banks in CEE countries concentrate more than 60% of their assets in loans. All other banks have low or medium values of the ratio of *loans to total assets*, which indicates that they invest prevailing shares of their funds in alternative assets. Furthermore, only for Estonia and Slovenia, the CEE countries with highest involvement of banks in credit supply, the distribution of the *loans to total assets* variable indicates levels of lending activity that are higher than those of Greek banks. Only in these two countries do the majority of banks invest predominantly (more than 60% of their assets) in loans. In Hungary and Poland the prevailing share of banks invests between 40% and 60% of their assets as loans, a feature that resembles the Greek banking sector. For all other CEE countries (Bulgaria, the Czech Republic, Latvia, Lithuania, Romania and Slovakia) the majority of banks invest less than 40% of their assets in loans. In these countries only very few banks distribute more than 60% of their assets as loans, which signals the importance of alternative assets in bank portfolios.

The results from this section confirm the hypothesis that not only the smaller size of the banking industry, but also the fact that existing banks invest in alternative assets instead of supplying credit, explains the low levels of bank loan supply. In order to illustrate the importance of the alternative investments in the asset portfolio of banks we present the distributions of the variables *assets in form of government securities to total assets* and *deposits with banks to total assets*.

The distribution of the variable *assets in form of government securities to total assets*, presented in Figures A.15. to A.23 in the Appendix, indicates that investment in government securities represents a substantial share of assets in Bulgaria (especially in the early transition period), Hungary, Lithuania, Poland and Slovakia²⁶. In Bulgaria, Hungary, Lithuania and Slovakia almost half of the banks, and in Poland even more than 80% of the banks, invest more than 10% of their assets in government securities. In the developed EU financial systems this is done by less than 5% of the banks.

²⁶ Data for bank investments in government securities in the Czech Republic, Estonia, Latvia and Romania is not available

All CEE countries, but Poland, have recently experienced periods of severe government deficits. These deficits have mostly been financed by government bonds which represented a profitable risk adjusted investment alternative for banks. Furthermore, all CEE countries had wide-ranging consolidation programs²⁷ that swapped bad bank loans for government securities. The effects of these consolidation programs can be seen in the high share of government securities in bank assets. The implementation of consolidation programs is marked by an increase in the share of banks that invest more than 10% of their assets in government securities (i.e. Hungary and Poland in 1994).

The literature on transition banking points to the possibility that the high share of government bonds in bank assets can generate crowding-out effects (Anderson and Kegels, 1998). The lack of loan demand data does not allow us to prove the relevance of this hypothesis. Nevertheless, an alternative argument can be mentioned here. In CEE countries, government debt is intermediated by the banking sectors, whereas in countries with stronger investment tradition the public directly holds government securities and these are not reflected on bank balance sheets. This means that banks in CEE countries undertake functions that are performed by capital markets in mature market economies. This implies that banks increase in size due to reasons not directly associated with typical banking functions like credit assessment and monitoring.

The distribution of *deposits with banks to total assets*, presented in Figures A.24 to A.35 in the Appendix, show that a large proportion of banks in Bulgaria, the Czech Republic, Poland, Romania and Slovakia hold a substantial share of assets in the form of deposits with other banks.

On the liability side, significant differences among the distribution of the variable *customer deposits to total deposits* across the EU benchmark countries is observed. In Germany and Greece most of the banks rely predominantly on funds gathered in the customer deposit market. In contrast, funding through the interbank market has a much higher importance for

²⁷ in the case of Hungary and Poland multiply repeated

Dutch banks²⁸. As reflected in Figures A.36 to A.48 in the Appendix we also observe significant differences among the CEE countries. In the Czech Republic, Hungary, Poland and Slovakia interbank deposits are an important source of refinancing for the vast majority of banks.

On the other hand, in Estonia, Latvia, Lithuania²⁹, Romania and Slovenia the distribution of the variable *customer deposits to total deposits* shows that almost all banks are fully financed by customer deposits. A low ratio of customer deposits to total deposits ratio is observed only temporary in periods of instability of the financial system (e.g. Latvia in 1995-1996, Romania in 1997-98). This stems from the fact that central bank liquidity assistance is reflected in the balance sheets as interbank deposit. The distributions of the variable customer deposits to total deposits for the CEE countries are otherwise similar to the ones of German and Greek banks. They indicate a relative low importance of interbank deposits on the liability side of bank balance sheets. In these CEE countries all banks participate in the deposit market under similar conditions and the incentives for interbank borrowing are limited. Bulgarian banks showed a relatively low involvement in deposit collecting activities during the mid-1990s, but later this pattern changes and by 2001 most banks finance their assets almost exclusively by customer deposits.

The distribution of the variables measuring financial intermediation productivity of banks point to both, differences across CEE countries, and differences across banks within the respective banking systems. These results comply with the specialization argument made by some studies on banking in CEE countries. Large incumbent banks in CEE countries specialize in deposit gathering, whereas loan supply is mostly provided by new entrant banks. This phenomenon has been described by Miller and Petranov (2002) for the early transition period in Bulgaria and by Bonin et. al. (1998) for the Czech Republic and Hungary. The

²⁸ As evidenced by De Bandt and Davis (1999) the low level of customer deposits in the Netherlands is a phenomenon of the 1990s. The reason is that bank deposits became less attractive during the 1990s due to the higher returns in investment in shares and mutual funds. This low level of customer deposits is compensated for by deposits from foreign banks.

²⁹ After the Baltic countries' became independent from the USSR the assets of the local Sberbank subsidiaries were never recovered from the legal successor of Sberbank in Moscow²⁹. This is analogous to the situation in Slovenia (Bonin, et. al., 1998).

existence of such specialization explains the high share of interbank borrowing by CEE banks, because banks clear the systematic inequalities between deposits and loans through interbank trade.

		1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	three largest banks	72	40	46	27	22	22	22	25
Bulgaria	small banks	41	35	13	21	30	36	37	39
Czoch Bopublic	three largest banks	47	45	46	48	44	37	29	29
Czech Republic	small banks	59	52	54	49	47	39	33	35
Ectonia	three largest banks	40	43	52	53	57	57	58	55
Estonia	small banks	41	45	48	48	61	50	48	55
Hungon	three largest banks	33	31	31	33	34	35	43	47
Hungary	small banks	41	40	39	41	39	46	53	53
Lotvia	three largest banks	36	27	34	37	51	49	50	56
Latvia	small banks	28	23	17	21	39	38	27	34
Lithuania	three largest banks	54	59	50	42	40	45	40	41
Lithuania	small banks	55	46	34	33	44	46	48	43
Boland	three largest banks	24	24	25	28	30	42	44	40
Poland	small banks	33	39	44	47	49	52	49	49
Bomonia	three largest banks	35	46	51	43	31	23	24	29
Romania	small banks	60	56	56	38	35	31	35	37
Clovakia	three largest banks	45	41	44	43	46	47	43	22
SIOVANIA	small banks	66	59	56	48	41	48	43	49
Slovenia	three largest banks	52	37	39	41	47	51	51	47
Slovenia	small banks	43	53	46	44	49	55	54	52

Table 13: Loans to total assets, weighted average by groups of banks, in %

Source: Own calculations based on BankScope data

To present more evidence supporting this phenomenon we look at the differences among the financial intermediation performance of different groups of banks. We divide the sample of banks covered by BankScope into two groups. The first includes the three largest banking institution in terms of total assets, and the second includes the remaining "small" banks. Table A.2 and A.3 in the Appendix illustrate that in Bulgaria, the Czech Republic, Hungary, Poland, and Slovakia all large banks are incumbent institutions. In newly independent economies like Estonia, Latvia, Lithuania and Slovenia, in contrast, incumbent banks did not exist and new entrants managed to acquire a dominant position in the banking market.

Table 13 and Table 14 illustrate that the largest incumbent banking institutions do indeed perform different activities when compared with the rest of the banking sectors in Bulgaria, the Czech Republic, Hungary, Poland and Slovakia. So for example, the ratio of loans to total assets in the Czech Republic and Hungary is on average 10% higher for small banks than for the large banks. In Poland and Slovakia small banks have a ratio of loans to total assets that is even about 20% higher than the one for the largest banks. On the liability side, large banks in

the Czech Republic, Hungary, Poland and Slovakia almost exclusively finance their activities by customer deposits, whereas customer deposits represent only 50-70% of the total deposit in small banks. Thus up to 50% of small banks' deposits represent interbank financing.

These differences cannot be explained by the size of the banks alone, since in countries where dominant banks are new entrant institutions (Estonia, Latvia, Lithuania, and Slovenia) this phenomenon of specialization is not observed.

	·	1994	1995	1996	1997	1998	1999	2000	2001
Bulgaria	three largest banks	24	66	53	85	91	95	98	98
Биідагіа	small banks	74	89	88	89	88	89	88	87
Crech Bonublic	three largest banks	83	80	80	80	81	86	91	91
Czech Republic	small banks	59	62	65	65	73	73	59	63
Estania	three largest banks	96	86	83	75	79	81	88	92
Estonia	small banks	90	73	76	72	59	69	83	84
Hungon	three largest banks	85	86	87	88	86	91	89	92
Hungary	small banks	57	67	67	62	63	69	68	69
Lotvia	three largest banks	87	98	95	86	76	78	86	83
Latvia	small banks	88	89	82	86	79	81	84	84
Lithuania	three largest banks	84	92	91	88	83	84	86	87
Lithuania	small banks	81	83	73	86	78	85	80	75
Boland	three largest banks	92	95	94	94	100	93	91	91
Poland	small banks	73	71	67	70	72	72	71	78
Bomonio	three largest banks	59	50	51	60	86	86	90	91
Romania	small banks	78	84	92	89	85	83	86	89
Slovakia	three largest banks	72	81	84	86	87	88	93	93
SIOVAKIA	small banks	48	53	50	50	54	58	61	80
Clavania	three largest banks	83	86	87	89	86	84	84	82
Slovenia	small banks	89	89	88	90	90	88	87	85

Table 14: Customer deposits to total deposits, weighted average by groups of banks, in %

Source: Own calculations based on BankScope data

In general the small banks in those CEE sample countries where incumbents still dominate the deposit market rely heavily on interbank funds. This supports the notion of specialization within the banking sector. The process of channeling funds from savers to investors in the economy contains an additional chain of transferring funds from a few banks possessing access to a wide range of customers to those banks that have not developed a customer deposit network, but have potential projects to finance.

One immediate result of the specialization phenomenon is that the average ratio measuring financial intermediation productivity of banks is relatively low, since some banks are not involved in loan supply, whereas others do not gather deposits.

To summarize, the micro level analysis on the financial intermediation productivity of CEE banks indicates that the low aggregate levels of bank intermediation are a reflection of the low financial intermediation productivity. This is especially true with regard to the loan supply function; banks in all CEE countries are less involved in loan supply than banks from the EU benchmark countries. Moreover, a phenomenon of specialization of large incumbents in deposit gathering and of new entrants in loan supply is observed in some of the CEE countries (Bulgaria, the Czech Republic, Hungary, Poland and Slovakia). As a result ratios measuring the aggregate financial intermediation productivity of banks are biased downwards, since some banks are not involved in loan supply, whereas others do not gather deposits.

6. Conclusion

This paper presents an analysis of the size of the banking sectors in Central and Eastern European (CEE) countries. We focus on the banking sectors' ability to provide financial intermediation between savers and investors in the economy.

The existing literature on banking in transition economies argues in unison that banking sectors in CEE countries are too small and do not provide sufficient levels of financial intermediation. In this paper we detect a common drawback of the existing measures used to indicate the size of CEE banking sectors: they all relate the volume of bank intermediation to GDP. We argue that since transition economies have a low stock of financial wealth relative to economic activity, a more objective measure of the size of the banking sector is the ratio of bank assets to a proxy of the stock of financial wealth rather than to GDP. Indeed, we find evidence that the estimation of the size of the banking sectors relative to GDP produce downward biased measures for the ability of CEE banks to intermediate available financial wealth, a gap between the developed EU banking systems and those of the CEE countries exists. But this gap is not as severe as argued in studies based on the traditional approach of measuring the size of the banking system with respect to GDP.

This paper, moreover, focuses on three arguments proposed in the literature that help explain the low level of financial intermediation by banks in CEE countries.

First, we find that the low current level of bank intermediation can at least partly be explained by the low initial stage of development of CEE banking sectors. By comparing the levels of marginal intermediation of the CEE countries with those of the EU countries, we conclude that catching up with the EU levels is progressing, at least with regard to the CEE banks' ability to accumulate deposits. Regarding to the lending function of the banking sectors, a trend towards approaching EU levels could be observed during the early transition years but was interrupted in the late 1990s.

Second, comparing the capital markets of CEE countries with those of the EU benchmarks, we reject the hypothesis that the development of banking sectors has been insufficient because financial systems are developing in a market-based manner. The data illustrate that capital markets in all CEE countries are even more severely underdeveloped than banking sectors. Therefore, they cannot compensate for the low level of bank financial intermediation.

Finally, when analyzing the involvement of CEE banks in classical financial intermediation activities on a micro level we find evidence for the hypothesis that the low level of financial intermediation provided by CEE banking sectors is a reflection of the low financial intermediation productivity of CEE banks on the individual bank level. Furthermore, our data indicate that significant differences across banks in some of the CEE countries exist. Whereas some banks do not perform credit, others show only minor involvement in deposit gathering. We relate this heterogeneity to a phenomenon of bank specialization in some CEE countries that has previously been described in the literature. A comparison between the financial intermediation activities of the largest banks with those of the small institutions supports the argument that in countries where incumbent banks still dominate the deposit market, there is a specialization of large incumbents in deposit gathering and of new entrant banks in credit supply.

Bibliography:

Allen, F. and D. Gale. 1999. Comparing Financial Systems, MIT

- Anderson, R. and C. Kegels. 1998. Transition Banking: Financial Development of Central and Eastern Europe, *Oxford University Press*
- Barisitz, S. 2001. The Development of the Romanian and Bulgarian Banking Sectors since 1990, *Focus on Transition, Österreichische Nationalbank*
- Baxter, M. and M. Cruccini. 1993. Explaining Saving-Investment Correlations. *American Economic Review* 83: 416-36
- Bayoumi, T. 1990. Saving-Investment Correlations: Immobile Capital, Government Policy, or Endogenous Behaviour? *IMF Staff Papers* 37: 360-387
- Bonin, J. 2001. Financial Intermediation in Southeast Europe: Banking on the Balkans, Vienna Institute for International Economic Studies, mimeo
- Bonin, J., Mizsei, K., Szekely, I., and P. Wachtel. 1998. Banking in Transition Economies: Developing Market Oriented Sectors in Eastern Europe, *Institute for EastWest Studies*
- Bonin, J. and P. Wachtel. 2002. Financial Sector Development in Transition Economies: Lessons from the First Decade, *BOFIT Discussion Paper* 9/2002
- Calveras, A. 2001. Bank Specialization: The Role of the Interbank Market. mimeo
- Clarke, G., Cull, R., Peria, M. S., and S. M. Sanchez. 2003. Foreign Bank Entry: Experience, Implications for Developing Economies, and Agenda for Further Research. *The World Bank Research Observer*, vol. 18 (1)
- Cole, D., and B. Slade. 1996. Building a Modern Financial System: The Indonesian Experience. *Cambridge University Press*
- Danthine, J-P., Giavazzi, F., Vives, X., and E-L. von Thadden. 1999. The Future of European Banking, *Monitoring European Integration, CEPR*
- De Bandt, O. and E.P. Davis. 1999. A Cross-Country Comparison of Market Structures in European Banking, ECB Working Paper
- Demirgüc-Kunt, A. and R. Levine. 2001. Bank-based and Market-based Financial Systems. Financial Structure and Economic Growth, *MIT Press*
- Dittus, P. and S. Prowse. 1995. Corporate Control in Central Europe and Russia: Should Banks Own Shares? *World Bank*

European Central Bank. 2002. Financial Sectors in the EU Accession Countries

- Feldstein, M. And C. Horioka. 1980. Domestic Saving and Internatioanl Capital Flows. *The Economic Journal* 90: 314-329
- Fender, I. and J. von Hagen. 1998. Central Banking in a more Perfect Financial System. Open Economies Review Vol. 9: 331-353
- Folkerts-Landau, D., Mathieson, D. J. And G. J. Schinasi. 1997. International Capital Markets: Developments, Prospects and Key Policy Issues, *International Monetary Fund*
- Freixas, X. and J. C. Rochet. 1997. Microeconomics of Banking, MIT
- French, K. R. and J. M. Poterba. 1991. Investor Diversification and Internatioanal Equity Markets. *NBER Working Paper* 3609
- Grinblatt, M. and S. Titman. 1998. Financial Markets and Corporate Strategy, *McGraw-Hill* Grossman, R. S. 2001. Double Liability and Bank Risk Taking, *Journal of Money, Credit* and Banking 33(2): 143-159
- Hencsey, N. and G. Hultgren. 2001. Securities Markets: Huge Gap, Difficult to Close Soon. *EU Enlargement Monitor* Deutsche Bank Research 5/01
 Ho, S. Y., and A. Saunders. 1985. Micro Model of the Federal Funds Market. *The Journal of Finance* 49(3): 977-990
- Hristov, K. and M. Michailov. 2002. Bank Credit Activities and Credit Market Rationing in Bulgaria, *Bulgarian National Bank*
- Iakova, D. and N. Wagner. 2001. Financial Sector Evolution in Central European Economies: Challenges in Supporting Macroeconomic Stability and Sustainable Growth, *International Monetary Fund Working Paper*
- Koch-Weser, C. 1996. The Transition in Central and Eastern Europe: Recent Progress and Next Steps, Address to the Fourth Annual Conference "Banking and Finance: The Experience of Central Europe"
- Kwan, S. H. 2002. Operating Performance of Banks among Asian Economies: An International and Time Series Comparison, *Federal Reserve Bank of San Francisco*
- Miller, J. and S. Petranov. 2002. The Financial System in the Bulgarian Economy, *Bulgarian National Bank*
- Perotti, E. C. 1993. Bank Lending in Transition Economies, mimeo

- Petrov, B. 2000. Bank Reserve Dynamics under Currency Board Arrangement in Bulgaria. Bulgarian National Bank
- Pissarides, F. 2001. Financial Structures to Promote Private Sector Development in South-Eastern Europe, *European Bank for Development and Reconstruction Working Papers*
- Pyle, W. 2002. Over-banked and Credit-Starved: A Paradox of the Transition. mimeo
- Schardax, F. and T. Reiniger. 2001. The Financial Sector in Five Central and Eastern European Countries: An Overview, *Focus on Transition, Österreichische Nationalbank*
- Stulz, R. 2001. Does Financial Structure Matter for Economic Growth? Financial Structure and Economic Growth, *MIT Press*
- Transition Report .2001. European Bank for Recunstruction and Development
- Zoli, E. 2001. Cost and Effectiveness of Banking Sector Restructuring in Transition Economies, *International Monetary Fund*

Appendix

Table A.1: Data sources

Variable	Data Source
Bank assets to GDP	IFS
Bank claims to the private sector to GDP	IFS
Deposits with banks to GDP	IFS
Stock market capitalization to GDP	Transition Report for CEE countries, Datastream for EU benchmarks
Domestic debt securities to GDP	Bank for International Settlements
Loans to total assets	BankScope
Deposits with banks to total assets	BankScope
Investment in government securities to total assets	BankScope
Customer deposits to total deposits	BankScope

Bank	Incumbent	Customer deposit market share in							
Dailk	incumbent	1994	1995	1996	1997	1998	1999	2000	2001
BULBANK	yes	12%	12%	24%	28%	20%	27%	29%	30%
DSK	yes	45%	37%	24%	30%	28%	26%	25%	25%
United Bulgarian Bank	yes	14%	9%	6%	11%	10%	13%	15%	17%
Ceska Sporitelna	yes	39%	32%	26%	26%	26%	28%	29%	26%
Ceskoslovenska Obchodni Banka	yes	8%	9%	8%	10%	9%	11%	28%	28%
Komercni Banka	yes	27%	25%	23%	24%	20%	22%	23%	22%
Estonian Savings Bank	yes	35%	28%	23%	19%	-	-	-	-
Eesti Uhispank	no	36%	20%	22%	21%	29%	27%	23%	24%
HansaBank	no	10%	28%	32%	29%	60%	64%	68%	68%
Hungarian Foreign Trade Bank	yes	6%	6%	7%	8%	8%	8%	8%	9%
K&H Bank	yes	10%	9%	9%	9%	10%	14%	14%	14%
National Savings and Commercial Bank	yes	55%	49%	44%	39%	40%	34%	30%	31%
Hansabanka	no	5%	8%	12%	17%	20%	22%	22%	24%
Latvijas Unibanka	no	23%	25%	24%	24%	34%	31%	28%	25%
Parekss Banka	no	27%	22%	24%	24%	34%	37%	33%	28%
AB Bankas Hansa	no	-	26%	24%	30%	39%	43%	33%	33%
Agricultural Bank of Lithuania	yes	22%	20%	21%	20%	17%	17%	13%	13%
Commercial Bank of Lithuania	yes	46%	24%	20%	-	-	-	-	-
Vilniaus Bankas	no	18%	7%	15%	21%	21%	29%	41%	41%
Bank Pekao	yes	14%	13%	15%	15%	12%	13%	13%	15%
Bank Przemyslowo-Handlowy PBK	yes	4%	6%	5%	7%	8%	10%	11%	13%
Powszechna Kasa Oszczednosci BP	yes	24%	32%	30%	27%	34%	25%	28%	28%
Banca Comerciala Romana	no	8%	12%	12%	13%	28%	34%	29%	35%
BANCOREX	yes	79%	66%	71%	54%	23%	-	-	-
Romanian Savings Bank	yes	10%	8%	10%	11%	13%	14%	11%	-
Slovak Savings Bank	yes	49%	46%	43%	41%	36%	34%	32%	31%
Tatra Banka	no	3%	5%	7%	7%	7%	9%	12%	15%
Vseobecna Uverova Banka	yes	33%	32%	30%	29%	26%	26%	25%	25%
Nova Kreditna Banka Maribor	yes	22%	18%	13%	10%	16%	16%	18%	16%
Nova Ljubljanska Banka	no	23%	21%	20%	21%	46%	45%	48%	47%
SKB Banka DD	no	7%	10%	10%	9%	14%	13%	14%	14%

Table A.2: Main deposit gathering banks in CEE: shares in deposit market

Source: Own calculations based on Bankscope and IFS

Bonk	Incumbont	Share in total assets of the banking system							
Bank	incumpent	1994	1995	1996	1997	1998	1999	2000	2001
BULBANK	yes	29%	27%	40%	36%	30%	30%	27%	25%
DSK	yes	20%	21%	14%	12%	13%	14%	14%	15%
United Bulgarian Bank	yes	8%	7%	8%	11%	12%	13%	13%	15%
Ceska Sporitelna	yes	20%	20%	18%	19%	19%	19%	20%	20%
Ceskoslovenska Obchodni Banka	yes	12%	12%	11%	12%	12%	13%	23%	23%
Komercni Banka	yes	22%	23%	22%	22%	19%	19%	19%	18%
Estonian Savings Bank	yes	21%	21%	18%	18%	-	-	-	-
Eesti Uhispank	no	23%	17%	16%	20%	25%	23%	17%	17%
HansaBank	no	8%	23%	30%	30%	58%	58%	61%	65%
Hungarian Foreign Trade Bank	yes	9%	9%	9%	8%	10%	9%	10%	10%
K&H Bank	yes	11%	10%	9%	8%	8%	13%	13%	12%
National Savings and Commercial Bank	yes	31%	30%	30%	26%	26%	24%	23%	23%
Hansabanka	no	6%	9%	13%	19%	23%	24%	24%	26%
Latvijas Unibanka	no	21%	23%	23%	22%	28%	26%	26%	26%
Parekss Banka	no	26%	25%	25%	23%	28%	33%	31%	30%
AB Bankas Hansa	no	-	28%	27%	29%	34%	38%	40%	39%
Agricultural Bank of Lithuania	yes	20%	18%	18%	18%	16%	16%	12%	11%
Commercial Bank of Lithuania	yes	42%	23%	19%	-	-	-	-	-
Vilniaus Bankas	no	20%	17%	16%	18%	19%	19%	41%	41%
Bank Pekao	yes	16%	14%	12%	12%	17%	18%	17%	18%
Bank Przemyslowo-Handlowy PBK	yes	6%	8%	7%	7%	8%	13%	15%	15%
Powszechna Kasa Oszczednosci BP	yes	22%	20%	20%	20%	24%	19%	20%	21%
Banca Comerciala Romana	no	10%	11%	12%	12%	24%	31%	29%	32%
BANCOREX	yes	76%	62%	60%	52%	21%	-	-	-
Romanian Savings Bank	yes	8%	7%	7%	7%	8%	10%	9%	-
Slovak Savings Bank	yes	41%	41%	39%	36%	32%	32%	27%	27%
Tatra Banka	no	3%	6%	8%	7%	8%	9%	13%	16%
Vseobecna Uverova Banka	yes	30%	29%	29%	26%	23%	24%	22%	22%
Nova Kreditna Banka Maribor	yes	22%	18%	13%	9%	15%	14%	16%	18%
Nova Ljubljanska Banka	no	22%	22%	22%	21%	40%	45%	48%	42%
SKB Banka DD	no	8%	9%	8%	9%	13%	13%	14%	13%

Table A.3: Main deposit gathering banks in CEE: shares in total assets of the banking system

Source: Own calculations based on Bankscope and IFS

Figure A.1: Structure of bank balance sheet in BankScope





Figure A.2: Distribution of variable loans to total assets for the Bulgarian banks, 1995-2001



Figure A.3: Distribution of variable loans to total assets for the Czech banks, 1995-2001



Figure A.4: Distribution of variable loans to total assets for the Estonian banks, 1993-2001



Figure A.5: Distribution of variable loans to total assets for the Hungarian banks, 1993-2001



Figure A.6: Distribution of variable loans to total assets for the Latvian banks, 1993-2001



Figure A.7: Distribution of variable loans to total assets for the Lithuanian banks, 1993-2001



Figure A.8: Distribution of variable loans to total assets for the Polish banks, 1993-2001



Figure A.9: Distribution of variable loans to total assets for the Romanian banks, 1993-2001



Figure A.10: Distribution of variable loans to total assets for the Slovakian banks, 1993-2001



Figure A.11: Distribution of variable loans to total assets for the Slovenian banks, 1993-2001



Figure A.12: Distribution of variable loans to total assets for the German banks, 1993-2001



Figure A.13: Distribution of variable loans to total assets for the Greek banks, 1993-2001



Figure A.14: Distribution of variable loans to total assets for the Dutch banks, 1993-2001



Figure A.15: Distribution of variable government securities to total assets for Bulgarian banks, 1993-2001



Figure A.16: Distribution of variable government securities to total assets for Hungarian banks, 1993-2001



Figure A.17: Distribution of variable government securities to total assets for Lithuanian banks, 1993-2001

Figure A. 18: Distribution of variable government securities to total assets for Polish banks, 1993-2001





Figure A.19: Distribution of variable government securities to total assets for Slovakian banks, 1993-2001

Figure A.20: Distribution of variable government securities to total assets for Slovenian banks, 1993-2001





Figure A.21: Distribution of variable government securities to total assets for German banks, 1993-2001

Figure A.22: Distribution of variable government securities to total assets for Greek banks, 1993-2001





Figure A.23: Distribution of variable government securities to total assets for Dutch banks, 1993-2001



Figure A.24: Distribution of variable deposits with banks to total assets for Bulgarian banks, 1993-2001

Figure A.25: Distribution of variable deposits with banks to total assets for Czech banks, 1993-2001





Figure A.26: Distribution of variable deposits with banks to total assets for Estonian banks, 1993-2001



Figure A.27: Distribution of variable deposits with banks to total assets for Hungarian banks, 1993-2001



Figure A.28: Distribution of variable deposits with banks to total assets for Latvian banks, 1993-2001



Figure A.29: Distribution of variable deposits with banks to total assets for Lithuanian banks, 1993-2001



Figure A.30: Distribution of variable deposits with banks to total assets for Polish banks, 1993-2001



Figure A.31: Distribution of variable deposits with banks to total assets for Romanian banks, 1993-2001



Figure A.32: Distribution of variable deposits with banks to total assets for Slovakian banks, 1993-2001



Figure A.33: Distribution of variable deposits with banks to total assets for Slovenian banks, 1993-2001



Figure A.34: Distribution of variable deposits with banks to total assets for German banks, 1993-2001



Figure A.35: Distribution of variable deposits with banks to total assets for Dutch banks, 1993-2001



Figure A.36: Distribution of variable customer deposits to total deposits for Bulgarian banks, 1993-2001



Figure A.37: Distribution of variable customer deposits to total deposits for Czech banks, 1993-2001



Figure A.38: Distribution of variable customer deposits to total deposits for Estonian banks, 1993-2001



Figure A.39: Distribution of variable customer deposits to total deposits for Hungarian banks, 1993-2001



Figure A.40: Distribution of variable customer deposits to total deposits for Latvian banks, 1993-2001



Figure A.41: Distribution of variable customer deposits to total deposits for Lithuanian banks, 1993-2001



Figure A.42: Distribution of variable customer deposits to total deposits for Polish banks, 1993-2001



Figure A.43: Distribution of variable customer deposits to total deposits for Romanian banks, 1993-2001



Figure A..44: Distribution of variable customer deposits to total deposits for Slovakian banks, 1993-2001



Figure A.45: Distribution of variable customer deposits to total deposits for Slovenian banks, 1993-2001



Figure A.46: Distribution of variable customer deposits to total deposits for German banks, 1993-2001



Figure A.47: Distribution of variable customer deposits to total deposits for Greek banks, 1993-2001



Figure A.48: Distribution of variable customer deposits to total deposits for Dutch banks, 1993-2001

ZEI Papers

PP = Policy Paper	WP = Working Paper
PP B97-01	A Stability Pact for Europe (a Forum organized by ZEI)
PP B97-02	Employment and EMU (Deutsch-Französisches Wirtschaftspolitisches Forum/ Forum Economique Franco-Allemand)
PP B97-03	Liberalising European Markets for Energy and Telecommunications: Some Lessons from the US Electric Utility Industry (Tom Lyon and John Mayo)
PP B97-04	Macroeconomic Stabilization with a Common Currency: Does European Monetary Unification Create a Need for Fiscal Insurance or Federalism? (Kenneth Kletzer)
PP B98-01	Budgeting Institutions for Aggregate Fiscal Discipline (Jürgen von Hagen)
PP B98-02	Trade with Low-Wage Countries and Wage Inequality (Jaleel Ahmad)
PP B98-03	Central Bank Policy in a More Perfect Financial System (Jürgen von Hagen and Ingo Fender)
PP B98-04	The EMU's Exchange Rate Policy (Deutsch-Französisches Wirtschaftspolitisches Forum/ Forum Economique Franco-Allemand)
PP B98-05	Estimating a European Demand for Money (Bernd Hayo)
PP B98-06	Monetary Union, Asymmetric Productivity Shocks and Fiscal Insurance: an Analytical Discussion of Welfare Issues (Kenneth M. Kletzer)
PP B98-07	Designing Voluntary Environmental Agreements in Europe: Some Lessons from the U.S. EPA's 33/50 Program (John W. Maxwell)
WP B98-08	Money-Output Granger Causality Revisited: An Empirical Analysis of EU Countries (Bernd Hayo)
WP B98-09	US Monetary Policy AND Monetary Policy and the ESCB (Robert L. Hetzel)
PP B98-10	Der Wettbewerb der Rechts- und politischen Systeme in der Europäischen Union (Martin Seidel)
WP B98-11	Exchange Rate Regimes in the Transition Economies: Case Study of the Czech Republic: 1990-1997 (Julius Horvath)
PP B98-11A	Die Bewertung der " dauerhaft tragbaren öffentlichen Finanzlage" der EU Mitgliedstaaten beim Übergang zur dritten Stufe der EWWU (Rolf Strauch)
WP B98-12	Price Stability and Monetary Policy Effectiveness when Nominal Interest Rates are Bounded at Zero (Athanasios Orphanides and Volker Wieland)
WP B98-13	Fiscal Policy and Intranational Risk-Sharing (Jürgen von Hagen)
WP B98-14	Free Trade and Arms Races: Some Thoughts Regarding EU-Russian Trade (Rafael Reuveny and John Maxwell)
WP B98-15	Can Taxing Foreign Competition Harm the Domestic Industry? (Stefan Lutz)
PP B98-16	Labour Market & Tax Policy in the EMU (Deutsch-Französisches Wirtschaftspolitisches Forum/ Forum Economique Franco-Allemand)
WP B99-01	The Excess Volatility of Foreign Exchange Rates: Statistical Puzzle or Theoretical Artifact? (Robert B.H. Hauswald)
WP B99-02	The Consequences of Labour Market Flexibility: Panel Evidence Based on Survey Data (Rafael Di Tella and Robert MacCulloch)
WP B99-03	The Macroeconomics of Happiness (Rafael Di Tella, Robert MacCulloch and Andrew J. Oswald)
WP B99-04	The Finance-Investment Link in a Transition Economy: Evidence for Poland from Panel Data (Christian Weller)
WP B99-05	Tumbling Giant: Germany's Experience with the Maastricht Fiscal Criteria (Jürgen von Hagen and Rolf Strauch)
WP B99-06	Productivity Convergence and Economic Growth: A Frontier Production Function Approach (Christopher M. Cornwell and Jens-Uwe Wächter)
WP B99-07	Comovement and Catch-up in Productivity Across Sectors: Evidence from the OECD (Christopher M. Cornwell and Jens-Uwe Wächter)
WP B99-08	The Connection Between More Multinational Banks and Less Real Credit in Transition Economies (Christian Weller)
WP B99-09	Monetary Policy, Parameter Uncertainty and Optimal Learning (Volker Wieland)

You can download these papers at the ZEI web site www.zei.de/zei_english/publikation/publ_zeib_pp.htm

WP B99-10	Financial Liberalization, Multinational Banks and Credit Supply: the Case of Poland (Christian Weller)
PP B99-11	Financial Supervision and Policy Coordination in the EMU (Deutsch- Französisches Wirtschaftspolitisches Forum / Forum Economique Franco- Allemand)
WP B99-12	Size Distortions of Tests of the Null Hypothesis of Stationarity: Evidence
WP B99-13	Financial Fragility or What Went Right and What Could Go Wrong in Central European Banking? (Christian E. Weller and Jürgen von Hagen)
WP B99-14	Industry Effects of Monetary Policy in Germany (Bernd Hayo and Birgit
WP B99-15	Financial Crises after Financial Liberalization: Exceptional Circumstances or Structural Weakness? (Christian F. Weller)
WP B99-16	Multinational Banks and Development Finance (Christian E. Weller and Mark J. Scher)
WP B99-17	Stability of Monetary Unions: Lessons from the Break-up of Czechoslovakia (Jan Fidrmuc, Julius Horvath and Jarko Fidrmuc)
WP B99-18	Why are Eastern Europe's Banks not failing when everybody else's are? (Christian E. Weller and Bernard Morzuch)
WP B99-19	The Evolution of Monetary Policy in Transition Economies (Ali M. Kutan and Josef C. Brada)
WP B99-20	Subnational Government Bailouts in Germany (Helmut Seitz)
WP B99-21	The End of Moderate Inflation in Three Transition Economies? (Josef C. Brada and Ali M. Kutan)
WP B99-22	Partisan Social Happiness (Rafael Di Tella and Robert MacCulloch)
WP B99-23	Informal Family Insurance and the Design of the Welfare State (Rafael Di Tella and Robert MacCulloch)
WP B99-24	What Makes a Revolution? (Robert MacCulloch)
WP B99-25	Micro and Macro Determinants of Public Support for Market Reforms in Eastern Europe (Bernd Hayo)
WP B99-26	Skills, Labour Costs, and Vertically Differentiated Industries: a General Equilibrium Analysis (Stefan Lutz and Alessandro Turrini)
WP B00-01	Monetary Union and Fiscal Federalism (Kenneth Kletzer and Jürgen von Hagen)
WP B00-02	Inflation Bias and Productivity Shocks in Transition Economies: The Case of the Czech Republic (Josef C. Brada, Arthur E. King and Ali M. Kutan)
WP B00-03	Integration, Disintegration and Trade in Europe: Evolution of Trade Relations During the 1990's (Jarko Fidrmuc and Jan Fidrmuc)
PP B00-04	A New Political Culture in the EU – Democratic Accountability of the ECB (Christa Randzio-Plath)
WP B00-05	Liberalization, Democracy and Economic Performance during Transition (Jan Fidrmuc)
PP B00-06 PP B00-07	EMU and Economic Growth in Europe (Deutsch-Französisches
WP B00-08	The Effectiveness of Self-Protection Policies for Safeguarding Emerging Market Economies from Crises (Kenneth Kletzer)
WP B00-09	Rational Institutions Yield Hysteresis (Rafael Di Tella and Robert MacCulloch)
WP B00-10	The Importance of Domestic Political Institutions: Why and How Belgium and Italy gualified for EMU (Mark Hallerberg)
WP B00-11	A Dynamic Approach to Inflation Targeting in Transition Economies (Lucjan T. Orlowski)
PP B00-12	Rechtsetzung und Rechtsangleichung als Folge der einheitlichen europäischen Währung (Martin Seidel)
WP B00-13	Back to the Future: The Growth Prospects of Transition Economies Reconsidered (Nauro F. Campos)
WP B00-14	Sources of Real Exchange Rate Fluctuations in Transition Economies: The Case of Poland and Hungary (Selahattin Dibooglu and Ali M. Kutan)

You can download these papers at the ZEI web site www.zei.de/zei_english/publikation/publ_zeib_pp.htm

WP B00-15	Regional Risksharing and Redistribution in the German Federation (Jürgen von Hagen and Ralf Hepp)
PP B00-16	The European Central Bank: Independence and Accountability (Christa
PP B00-17	Randzio-Plath and Tomasso Padoa-Schioppa) Rückführung der Landwirtschaftspolitik in die Verantwortung der Mitgliedsstaaten? – Rechts- und Verfassungsfragen des Gemeinschaftsrechts (Martin Seidel)
WP B00-18	Budget Processes: Theory and Experimental Evidence (Karl-Martin Ehrhart, Roy Gardner, Jürgen v. Hagen and Claudia Keser)
WP B00-19	Income Dynamics and Stability in the Transition Process – General Reflections applied to the Czech Republic (Jens Hölscher)
WP B00-20	Breaking-Up a Nation, from the Inside (Etienne Farvaque)
WP B01-01	Divided Boards: Partisanship through Delegated Monetary Policy (Etienne Farvaque, Gaël Lagadec)
WP B01-02	The Konstanz Seminar on Monetary Theory and Policy at Thirty (Michele
WP B01-03	Preferences over Inflation and Unemployment: Evidence from Surveys of Happiness (Rafael di Tella Robert J. MacCulloch and Andrew J. Oswald)
WP B01-04	The Determination of Umemployment Benefits (Rafael di Tella and Robert J. MacCulloch)
PP B01-05	Trade Rules and Global Governance: A Long Term Agenda / The Future of Banking (Deutsch-Französisches Wirtschaftspolitisches Forum/ Forum Economique Franco-Allemand)
WP B01-06	Opposites Attract: The Case of Greek and Turkish Financial Markets (Konstantinos Drakos and Ali M Kutan)
WP B01-07	The Convergence of Monetary Policy between Candidate Countries and the European Union (Josef C. Brada and Ali M. Kutan)
WP B01-08	The Functioning of Economic Policy Coordination (Jürgen von Hagen and Susanne Mundschenk)
WP B01-09	Democracy in Transition Economies: Grease or Sand in the Wheels of Growth? (Jan Fidrmuc)
WP B01-10	Integration of the Baltic States into the EU and Institutions of Fiscal Convergence: A Critical Evaluation of Key Issues and Empirical Evidence (Ali M Kutan and Nina Bautola Mol)
WP B01-11	Inflationary Performance in a Monetary Union with Large Wage Setters (Lilia Cavallari)
PP B01-12	The Impact of Eastern Enlargement on EU-Labour Markets / Pensions Reform Between Economic and Political Problems (Deutsch-Französisches Wirtschaftspolitisches Forum/Forum Economigue Franco-Allemand)
WP B01-13	German Public Finances: Recent Experiences and Future Challenges (Jürgen von Hagen und Rolf R. Strauch)
WP B01-14	Formal Fiscal Restraints and Budget Processes as Solutions to a Deficit and Spending Bias in Public Finances – U.S. Experience and Possible Lessons for FMU (Rolf Strauch and Jürgen von Hagen)
WP B01-15	Programs without Alternative: Public Pensions in the OECD (Christian E. Weller)
WP B01-16	Sources of Inflation and Output Fluctuations in Poland and Hungary: Implications for Full Membership in the European Union (Selahattin Dibooglu and Ali M Kutan)
WP B01-17	Executive Authority, the Personal Vote, and Budget Discipline in Latin American and Carribean Countries (Mark Hallerberg and Patrick Marier)
WP B01-18	Monetary Policy in Unknown Territory The European Central Bank in the Early Years (Jürgen von Hagen and Matthias Brückner)
WP B01-19	Detrending and the Money-Output Link: International Evidence (R.W. Hafer and Ali M. Kutan)
WP B01-20	An Empirical Inquiry of the Efficiency of Intergovernmental Transfers for Water Projects based on the WRDA Data (Anna Rubinchik-Pessach)

WP B01-21	Balkan and Mediterranean Candidates for European Union Membership: The Convergence of their Monetary Policy with that of the European Central Bank (Josef C. Brada and Ali M. Kutan)
	Dalik (JUSEI C. Diaua allu Ali IVI. Kulali) Stratagia Delegation and International Capital Taylation (Matthias Drückner)
VVP BU1-22	Strategic Delegation and International Capital Taxation (Matthas Bruckher)
WP B01-23	Migration and Adjustment to Snocks in Transition Economies (Jan Fidrmuc)
WP B01-24	Disintegration and Trade (Jarko and Jan Fidrmuc)
WP B01-25	Monetary Convergence of the EU Candidates to the Euro: A Theoretical Framework and Policy Implications (Lucjan T. Orlowski)
WP B01-26	Regional Effects of Terrorism on Tourism: Evidence from Three Mediterranean Countries (Konstantinos Drakos and Ali M. Kutan)
WP B01-27	Investor Panic, IMF Actions, and Emerging Stock Market Returns and Volatility: A Panel Investigation (Bernd Havo and Ali M Kutan)
PP B01-28	Political Economy of the Nice Treaty: Rebalancing the EU Council / The Future of European Agricultural Policies (Forum Economique Franco- Allemand / Deutsch-Französisches Wirtschaftspolitisches Forum)
WP 01-29	Is Kazakhstan vulnerable to the Dutch Disease? (Karlygash Karalbayeva, Ali M. Kutan and Michael L. Wyzan)
WP B02-01	Does Inflation Targeting Matter? (Manfred J.M. Neumann and Jürgen von Hagen)
WP B02-02	The Euro System and the Federal Reserve System Compared: Facts and Challenges (Karlbeinz Ruckriegel and Franz Seitz)
WP B02-03	The Choice of Exchange Rate Systems: An Empirical Analysis for Transition Economies (Jürgen von Hagen and Jizhong Zhou)
WP B02-04	Asymmetric Monetary Policy Effects in EMU (Volker Clausen and Bernd Havo)
WP B02-05	Real and Monetary Convergence Within the European Union and Between the European Union and Candidate Countries: A Rolling Cointegration
WP B02-06	Is there Asymmetry in Forward Exchange Rate Bias? Multi-Country Evidence (Su Zhou and Ali M. Kutan)
WP B02-07	Perspektiven der Erweiterung der Europäischen Union (Martin Seidel)
WP B02-08	Has the Link between the Spot and Forward Exchange Rates Broken Down? Evidence from Rolling Cointegration Tests (Ali M. Kutan and Su Zhou)
WP B02-09	Monetary Policy in the Euro Area – Lessons from the First Years (Volker Clausen and Bernd Havo)
PP B02-10	National Origins of European Law: Towards an Autonomous System of European Law? (Martin Seidel)
WP B02-11	The Eurosystem and the Art of Central Banking (Gunnar Heinsohn and Otto Steiger)
WP B02-12	Argentina: The Anatomy of a Crisis (Jiri Jonas)
WP B02-13	De Facto and Official Exchange Rate Regimes in Transition Economies (Jürgen von Hagen and Jizhong Zhou)
WP B02-14	The Long and Short of It: Global Liberalization, Poverty and Inequality (Christian F. Weller and Adam Hersh)
WP B02-15	Does Broad Money Matter for Interest Rate Policy? (Matthias Brückner and Andreas Schabert)
WP B02-16	Regional Specialization and Concentration of Industrial Activity in Accession Countries (Julia Traistary Peter Nijkamp and Simonetta Longhi)
WP B02-17	Specialization and Growth Patterns in Border Regions of Accession Countries (Laura Resmini)
WP B02-18	Regional Specialization and Employment Dynamics in Transition Countries (Iulia Traistaru and Guntram B. Wolff)
WP B02-19	East Germany: Transition with Unification, Experiments and Experiences (Jürgen von Hagen, Rolf R. Strauch and
	Guntram B. Wolff)
WP B02-20	The Impact of News, Oil Prices, and International Spillovers on Russian Financial Markets (Bernd Hayo and Ali M. Kutan)

WP B02-21	Nominal and Real Stochastic Convergence within the Transition Economies and to the European Union: Evidence from Panel Data (Ali M. Kutan and Taner M. Yigit)
WP B02-22	Der Staat als "Lender of Last Resort" – oder: Die Achillesferse des Eurosystems (Otto Steiger)
PP B02-23	Legal Aspects of European Economic and Monetary Union (Martin Seidel)
WP B02-24	The Effects of Quotas on Vertical Intra-Industry Trade (Stefan Lutz)
WP B02-25	Trade Policy: Institutional" vs. Economic" Factors (Stefan Lutz)
W/P B02-26	Monetary Convergence and Risk Premiums in the EU Candidate Countries
WE DO2-20	(Lucjan T. Orlowski)
WP B02-27	(Debajyoti Chakrabarty)
WP B02-28	Inequality, Politics and Economic Growth (Debajyoti Chakrabarty)
WP B02-29A	Growth and Business Cycles with Imperfect Credit Markets (Debajyoti Chakrabarty)
WP B02-29B	Trade Agreements as Self-Protection (Jennifer Pédussel Wu)
WP B02-30	An Adverse Selection Model of Optimal Unemployment Insurance (Marcus Hagedorn, Shok Kaul and Tim Mennel)
WP B03-01	Die Wirtschafts- und Währungsunion im rechtlichen und politischen Gefüge der Europäischen Union (Martin Seidel)
WP B03-02	Commuting in the Baltic States: Patterns, Determinants, and Gains (Mihails
	Hazans)
WP B03-03	Europäische Steuerkoordination und die Schweiz (Stefan H Lutz)
WP B03-04	Do Ukrainian Firms Benefit from FDI? (Stefan H Lutz and Oleksandr
	Talavera)
WP B03-05	Reconsidering the evidence: are Eurozone business cycles converging?
	(Michael Massmann and James Mitchell)
WP B03-06	Fiscal Discipline and Growth in Euroland
	Experiences with the Stability and Growth Pact
	(Jürgen von Hagen)
PP B03-07	Nach Nizza und Stockholm:
	Stand des Binnenmarktes und Prioritäten für die Zukunft (Martin Seidel)
WP B03-08	The Determination of Capital Controls: Which Role Do Exchange Rate
	Regimes Play? (Jürgen von Hagen and Jizhong Zhou)
WP B03-09	The European Central Bank and the Eurosystem:
	An Analysis of the Missing Central Monetary Institution in European
	Monetary Union (Gunnar Heinsohn and Otto Steiger)
WP B03-10	Foreign Direct Investment and Perceptions of Vulnerability to Foreign
	Exchange Crises: Evidence from Transition Economies (Josef C. Brada and
	Vladimír Tomsík)
PP B03-11	Die Weisungs- und Herrschaftsmacht der Europäischen Zentralbank im
	Europäischen System der Zentralbanken – eine rechtliche Analyse (Martin
	Seidel)
WP B03-12	What makes regions in Eastern Europe catching up? The role of foreign
	investment, human resources and geography (Gabriele Tondl and Goran
	Vuksic)
WP B03-13	The IS Curve and the Transmission of Monetary Policy: Is there a Puzzle?
	(Charles Goodhart and
	Boris Hofmann)
WP B03-14	FCIs and Economic Activity: Some International Evidence (Charles
	Goodhart and Boris Hofmann)
WP B03-15	Employed and unemployed search: the marginal willingness to pay for
	attributes in Lithuania, the US and the Netherlands (Jos van Ommeren and
	Mihails Hazans)
WP B03-16	South-East Europe: Economic Performance, Perspectives and Policy
	Challenges (Julia Traistary and Jürgen von Hagen)
WP B03-17	Determinants of inter-regional migration in the baltic countries (Mibails
	Hazans)
WP B03-18	The Effects of Regional and Industry-Wide FDI Spillovers on Export of

You can download these papers at the ZEI web site www.zei.de/zei_english/publikation/publ_zeib_pp.htm

WP B03-19	An Empirical Analysis of Competing Explanations of Urban Primacy. Evidence from Asia and the Americas (Ronald L. Moomaw and Mohammed A Alwosabi)
WP B03-20	Urban Primacy, Gigantism, and International Trade: Evidence from Asia and the Americas (Ronald L. Moomaw and Mohammed A. Alwosabe)
WP B03-21	Reputation Flows: Contractual Disputes and the Channels for Inter-firm
PP B03-22	Reformzwänge innerhalb der EU angesichts der Osterweiterung (Martin Seidel)
WP B03-23	Economic Integration and Manufacturing Concentration Patterns: Evidence
WP B03-24	Monetary Policy Reaction Functions: ECB versus Bundesbank (Bernd Hayo and Boris Hofmann)
WP B03-25	How Flexible are Wages in EU Accession Countries? (Anna Iara and Iulia Traistaru)
WP B03-26	Sovereign Risk Premia in the European Government Bond Market (Kerstin Bernoth, Juergen von Hagen and Ludger Schuknecht)
WP B03-27	The Performance of the Euribor Futures Market: Efficiency and the Impact of ECB Policy Announcements (Kerstin Bernoth and Juergen von Hagen)
WP B03-28	The Effects of Transition and Political Instability on Foreign Direct Investment: Central Europe and the Balkans (Josef C. Brada, Ali M. Kutan and Taner M. Yigit)
WP B03-29	Macroeconomic Implications of Low Inflation in the Euro Area (Jürgen von Hagen and Boris Hofmann)
PP B04-01	Die neuen Schutzklauseln der Artikel 38 und 39 des Beitrittsvertrages: Schutz der alten Mitgliedstaaten vor Störungen durch die neuen Mitgliedstaaten (Martin Seidel)
WP B04-02	Total Factor Productivity and Economic Freedom Implications for EU Enlargement (Ronald L. Moomaw and Euv-Seok Yang)
WP B04-03	Over- and underbidding in central bank open market operations conducted as fixed rate tender (Ulrich Bindseil)
WP B04-04	Who Is in Favor of Enlargement? Determinants of Support for EU Membership in the Candidate Countries' Referenda (Orla Doyle and Jan Fidrmuc)
WP B04-05 WP B04-06	Money Rules for the Eurozone Candidate Countries (Lucjan T. Orlowski) Rural-Urban Inequality in Africa: a Panel Study of the Effects of Trade Liberalization and Financial Deepening (Mina Baliamoune-Lutz and Stefan H. Lutz)
WP B04-07	The Contribution of Income, Social Capital, and Institutions to Human Well- Being in Africa (Mina Baliamoune-Lutz and Stefan H. Lutz)
WP B04-08	European Integration, Productivity Growth and Real Convergence (Taner M. Yigit and Ali M. Kutan)
WP B04-09	Testing Creditor Moral Hazard in Sovereign Bond Markets: A Unified Theoretical Approach and Empirical Evidence (Ayşe Y.Evrensel and Ali M. Kutan)
WP B04-10	Economic Integration and Industry Location in transition countries (Laura Resmini)
WP B04-11	Economic Integration and Location of Manufacturing Activities: Evidence from MERCOSUR (Pablo Sanguinetti, Iulia Traistaru and Christian Volpe Martineus)
WP B04-12	Measuring and Explaining Levels of Regional Economic Integration (Jennifer Pédussel Wu)
WP B04-13	The Role of Electoral and Party Systems in the Development of Fiscal Institutions in the Central and Eastern European Countries (Sami Yläoutinen)
WP B04-14 WP B04-15	Euro Adoption and Maastricht Criteria: Rules Or Discretion? (Jiri Jonas) Do Economic Integration and Fiscal Competition Help to Explain Location Patterns? (Christian Volpe Martincus)

WP B04-16	Does It Matter Where Immigrants Work? Traded Goods, Non-traded Goods, and Sector Specific Employment (Harry P. Bowen and Jennifer Pédussel Wu)
WP B04-17	Foreign Exchange Regime, the Real Exchange Rate and Current Account Sustainability: The Case of Turkey (Sübidey Togan and Hasan Ersel)
WP B04-18	Transmission Channels of Business Cycles Synchronization in an Enlarged EMU (Iulia Traistaru)
PP B04-19	Die Stellung der Europäischen Zentralbank nach dem Verfassungsvertrag (Martin Seidel)
WP B04-20	Money Market Pressure and the Determinants of Banking Crises (Jürgen von Hagen and Tai-kuang Ho)
WP B04-21	The effectiveness of subsidies revisited: accounting for wage and employment effects in business R&D (Volker Reinthaler and Guntram B. Wolff)
WP B04-22	Non-Discretionary Monetary Policy: The Answer for Transition Economies? (Elham Mafi-Kreft and Steven F. Kreft)
WP B04-23	Which Lender of Last Resort for the Eurosystem? (Otto Steiger)
WP B04-24	The Endogeneity of Money and the Eurosystem (Otto Steiger)
WP B04-25	Exchange Rate Risk and Convergence to the Euro (Lucian T. Orlowski)
WP B04-26	Analyzing Trade Opening in Likraine: Effects of a Customs Linion with the
WI D04-20	ELL (Okeene Herburyuk and Stofan Lutz)
	EU (Oksalia Halbuzyuk aliu Stelali Luiz)
VVP B04-27	
	(Galyna Grygorenko and Stefan Lutz)
WP B04-28	Fiscal Crises in U.S. Cities: Structural and Non-structural Causes
	(Guntram B. Wolff)
WP B04-29	Deutschlands Wirtschaft, seine Schulden und die Unzulänglichkeiten der
	einheitlichen Geldpolitik im Eurosystem
	(Dieter Spethmann and Otto Steiger)
WP B04-30	Der Vollzug von Gemeinschaftsrecht über die Mitgliedstaaten und seine
	Rolle für die FLL und den Beitrittsprozeß
	(Martin Seidel)
	The Effects of Transition and Political Instability on Eeroign Direct
VVF 004-31	Investment Inflows: Central Europe and the Delkane
	(least M. Drade, Ali M. Kutan and Tanan M. Visit)
	(Josef M. Brada, All M. Kutan and Taner M. Yigit)
WP B05-01	Macroeconomic Adjustment in the New EU Member States
	(Jürgen von Hagen and Iulia Traistaru)
WP B05-02	Voice of the Diaspora: An Analysis of Migrant Voting Behavior
	(Jan Fidrmuc and Orla Doyle)
WP B05-03	Public Education in an Integrated Europe: Studying to Migrate and
	Teaching to Stay?
	(Panu Poutvaara)
PP B05-04	Währungspolitik als Sozialpolitik
	(Martin Seidel)
WP B05-05	Regulatory Standards Can Lead to Predation
200 00	(Stefan Lutz)
WP 805-06	Banking Sector (Under?) Development in Central and Eastern Europe
	/ Jürgen von Hagen and Valeriva Dinger
	(Julyen von Hayen and Valenya Dingel)